Author’s response to reviews

Title: Serious postoperative complications induced by medical glue: three case reports

Authors:

Wei Li (weili8308@gmail.com; weili8308@126.com; weili8308@jlu.edu.cn)
Mingming Xiao (xia623245@163.com)
Yujia Chen (yujiachen129@163.com)
Jiaxing Yang (jiaxingyang@jlu.edu.cn)
Donghui Sun (sundonghui131@163.com)
Jian Suo (suojian0066@126.com)
Daguang Wang (daguangjida@163.com)

Version: 1 Date: 18 Oct 2018

Author’s response to reviews:

Dear Editor,

Thanks for your comments. Based on these comments, we have made careful revision.

1. CARE

   We have submitted the CAREchecklist as supplementary material in Attach Files.

2. Discussion and Conclusions

   We have combine "Discussion" and "Conclusions" sections into one "Discussion and Conclusions" section.

3. Authors Contributions

   The statement “All authors read and approved the final manuscript” have been added to the Authors’ contributions.
4. Declarations

"Declarations" heading have been added to the Declarations section.

5. Clean Manuscript

As suggested, we have checked the Manuscript.